Author's response to reviews

Title: First Episode of Psychosis and Migration in Italy (PEP_ITA Migration): a study in the Italian Mental Health Services

Authors:

Ilaria Tarricone (ilaria.tarricone@unibo.it)
Mauro Braca (mauro.braca@studio.unibo.it)
Fabio Allegri (fabio.allegri@studio.unibo.it)
Giuseppe Barrasso (barr.gius@libero.it)
Antonello Bellomo (a.bellomo@unifg.it)
Vanna Berlincioni (vanna.berlincioni@unipv.it)
Angelo Bertani (angelo.bertani@ao-sanpaolo.it)
Emanuele Caroppo (emanuele.caroppo@rm.unicatt.it)
Alessio Ceregato (alessio.ceregato@hotmail.it)
Paolo Cianconi (pcianco@gmail.com)
Marco Conforti Donati (m.conforti@systemidea.it)
Samuele Defilippi (csmchivasso@aslto4.piemonte.it)
Corrado De Rosa (corradoderosa@gmail.com)
Luigi Ferrannini (luigi.ferrannini@asl3.liguria.it)
Silvia Ferrari (silvia.ferrari@unimore.it)
Maria Antonietta Furio (marianto16@hotmail.it)
Carla Gramaglia (carla.gramaglia@gmail.com)
Caterina La Cascia (caterina.lacascia@unipa.it)
Mario Luciano (mario-luciano@hotmail.it)
Alice Mulè (alicemule15@yahoo.it)
Marcello Nardini (marcello.nardini@poste.it)
Francesca Podavini (francescapedavini@libero.it)
Diego Primavera (diego.primavera@tiscali.it)
Corrina Reginianini (corrina.regianini@alice.it)
Marco Rigatelli (marco.rigatelli@unimore.it)
Orlando Todarello (ortod@psichiat.uniba.it)
Elena Turella (etrella@ulss15.pd.it)
Simone Vender (simone.vender@uninsubria.it)
Antonio Ventriglio (a.ventriglio@libero.it)
Patrizia Zeppegno (patrizia.zeppegno@med.unipmn.it)
Domenico Berardi (domenico.berardi@unibo.it)
Andrea Fiorillo (anfioril@tin.it)

Version: 4 Date: 26 August 2013

Author's response to reviews: see over
Dear Editor,

We are pleased to submit you our paper “First Episode of Psychosis and Migration in Italy (PEP_ITA Migration): a study in the Italian Mental Health Services”. It is the protocol of a prospective observational study that is being carried out in 30 Italian mental health centers: to our knowledge, this is the first large-scale study about risk and protective factors for a First Episode Psychosis (FEP) in first generation migrants in Italy. We expect that the results of our study will allow a better understanding of the psychosis-causation process in first generation migrants in Italy, adding useful information to the literature available on this topic (mainly consisting in studies ruled out in only a few countries). Moreover, our results may contribute to the development of prevention programmes for psychosis and to improve early intervention treatments for the migrant population in Italy.

We hope you find this paper of interest for publication in your BMC Psychiatry. All authors agree on submission of the paper to BMC Psychiatry and confirm that they have no conflicting interests.

Thank you for your attention.

Sincerely,

Ilaria Tarricone